Author's response to reviews

Title: The Use and Abuse of Paracetamol in Children by Caregivers in Enugu, South-east, Nigeria

Authors:

Herbert Obu (herbert.obu@unn.edu.ng)
Josephat M chinawa (josephat.chinawa@unn.edu.ng)
Agozie C Ubesie (zionagoz@yahoo.co.uk)
Bismark C Eke (chriseke2006@yahoo.com)
Ikenna K Ndu (ikennandu@gmail.com)

Version: 2 Date: 9 March 2012

Author's response to reviews:

The editor
BMC Paediatrics

We humbly change the initials of the fifth author.he is Ikenna kingsaley Ndu and not Ndubuisi.Thanks